Author’s response to reviews

Title: Pro Inflammatory Interleukins and Thyroid Function in Naswar (Dipping Tobacco) Users: a case control study.

Authors:
Faiza Sajid (faizataha@gmail.com)
Samina Bano (samina_ku@hotmail.com)

Version: 2 Date: 23 Jul 2016

Author’s response to reviews:

Cover letter

Dear editor

Many thanks for reviewing our article titled Pro Inflammatory Interleukins and Thyroid Function in Naswar (Dipping Tobacco) Users: a case control study (BEND-D-16-00044). Following are the point-by-point responses to the comments.

Reviewer comment:
il-1 should be corrected as IL-1 on page 5 line 11

Response
Correction done.

Reviewer comment:
The correlation figures should be reduced only to the one/ones which are statistically significant.
Response

The correlation figure has been reduced to only the statistically significant one.

Thank you again & Best Regards

Faiza Sajid

Corresponding Author